Author's response to reviews

Title: Caution required when relying on a colleague's advice; a comparison between professional advice and evidence from the literature

Authors:

Frederieke G Schaafsma (f.g.schaafsma@amc.uva.nl)
Jos H Verbeek (jos.verbeek@ttl.fi)
Carel T Hulshof (c.t.hulshof@amc.uva.nl)
Frank J van Dijk (f.j.vandijk@amc.uva.nl)

Version: 4 Date: 20 June 2005

Author's response to reviews:

Dear Mrs Shaw,
Thank you for the news of acceptance of our paper.
We formatted the text anew according to the journals requirements.
I hope that the paper is ready for publication now.
With kind regards,
Jos Verbeek